Author's response to reviews

Title: Women's Opinions on Antenatal Care in Developing Countries: Results of a Study in Cuba, Thailand, Saudi Arabia and Argentina

Authors:

Dr Gustavo Nigenda (gnigenda@funsalud.org.mx)
Dr Ana Langer (alanger@popcouncil.org.mx)
Chusri Kuchaisit (CHUSRI-K@medlib2.kku.ac.th)
Mariana Romero (mromero@cedes.org)
Georgina Rojas (rojas@infomed.sld.cu)
Muneera Al-Osimy (mun16@hotmail.com)
Jose Villar (villari@who.ch)
Dr Jo Garcia (jo.garcia@perinat.ox.ac.uk)
Yagob Al-Mazrou (yalmazrou@hotmail.com)
Hassan Ba'aqeel (hassanbaaqeel@hotmail.com)
Guillermo Carrol (gcarroli@crep.org.ar)
Ubaldo Farnot (farnot@infomed.sld.cu)
Pisake Lumbiganon (pisake@kk1.kku.ac.th)
Jose Belizan (belizanj@clap.ops-oms.org)
Prof Per Bergsjo (p-bergsj@online.no)
Leiv Bakketeig (arosenkratz@health.sdu.dk)

Version: 5 Date: 2 May 2003


Editorial Board
BioMedCentral

Dear Editors:

You will find attached the final version of the paper "Women's Opinions on Antenatal Care in Developing Countries: Results of a Study in Cuba, Thailand, Saudi Arabia and Argentina", which contains the final changes that were suggested in your last message.

Sincerely yours

Gustavo Nigenda, PhD